



University of Groningen

Sexual identity disparities in mental health among U.K. adults, U.S. adults, and U.S. adolescents

Kiekens, Wouter J.; la Roi, Chaïm; Dijkstra, Jan Kornelis

Published in:

Psychology of Sexual Orientation and Gender Diversity

DOI:

10.1037/sgd0000432

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version
Publisher's PDF, also known as Version of record

Publication date: 2021

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA):

Kiekens, W. J., la Roi, C., & Dijkstra, J. K. (2021). Sexual identity disparities in mental health among U.K. adults, U.S. adults, and U.S. adolescents: Examining heterogeneity by race/ethnicity. *Psychology of Sexual Orientation and Gender Diversity*, *8*(4), 407-419. https://doi.org/10.1037/sgd00000432

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverne-amendment.

Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



© 2020 American Psychological Association ISSN: 2329-0382

2021, Vol. 8, No. 4, 407–419 http://dx.doi.org/10.1037/sgd0000432

Sexual Identity Disparities in Mental Health Among U.K. Adults, U.S. Adults, and U.S. Adolescents: Examining Heterogeneity by Race/Ethnicity

Wouter J. Kiekens University of Groningen Chaïm la Roi Institute for Future Studies, Stockholm, Sweden, and University of Stockholm

Jan Kornelis Dijkstra University of Groningen

Lesbian, gay, and bisexual (LGB) people report poorer mental health than heterosexual people. However, there is heterogeneity in this disparity, and a racial/ethnic minority identity can contribute to this heterogeneity. When studying the intersecting effect of sexual identity and race/ethnicity on mental health, research often limits race/ethnicity categories, often uses adult samples from the U.S., and often uses samples that are not nationally representative. To overcome these limitations, the present study examined racial/ethnic heterogeneity in mental health disparities between heterosexual and LGB people in three nationally representative samples. The samples used were the 2011-2012 Understanding Society (U.K. adults; N = 43,904), the 2015 National Survey on Drug Use and Health (U.S. adults; N = 43,313), and the 2015 Youth Risk Behavior Survey (U.S. adolescents; N = 15,122). Using these samples enabled us to contrast the intersection of sexual identity and race/ethnicity across countries (for adults), and between life phases (in the U.S.). Across all three samples, LGB people—and particularly bisexual people—had a higher risk of impaired mental health than heterosexual people. For U.K. adults and U.S. adults, no intersecting effect of sexual identity and race/ethnicity were found. LGB adolescents of color reported better mental health compared with White LGB adolescents. More specifically, Black LGB adolescents reported better mental health compared to White LGB adolescents. Together, the present study contributes to a better understanding of the heterogeneity in mental health disparities for LGB people.

Public Significance Statement

Mental health disparities for sexual minority people are found in both the U.S. and the U.K. Among U.S. adolescents, the negative effect of sexual identity was smaller among LGB adolescents of color and Black LGB adolescents specifically.

Keywords: mental health, sexual identity, race/ethnicity, adults, adolescents

Research has repeatedly found mental health disparities between lesbian, gay, and bisexual (LGB) and heterosexual people, with LGB people showing higher rates of mood and anxiety disorders and more substance use and suicide attempts than heterosexual people (King et al., 2008; Marshal, Dietz et al., 2011; Marshal, Friedman et al., 2008). Minority stress—which refers to the stressors uniquely experienced by LGB people, like heterosexist discrimination, internalized heterosexism, or concealment of

identity—has been proposed as an explanation for these health disparities (Hatzenbuehler, 2009; Meyer, 2003). However, some LGB people have worse mental health than others, and it is crucial to better understand this heterogeneity in mental health among LGB people to provide more specific support for these people.

One factor that can contribute to this heterogeneity is race/ ethnicity, which is understood as a social group or category de-

This article was published Online First August 6, 2020.

¹⁰ Wouter J. Kiekens, Department of Sociology, Interuniversity Center for Social Science Theory and Methodology (ICS), University of Groningen; ¹⁰ Chaïm la Roi, Institute for Future Studies, Stockholm, Sweden, and Swedish Institute for Social Research, University of Stockholm; Jan Kornelis Dijkstra, Department of Sociology, Interuniversity Center for Social Science Theory and Methodology (ICS), University of Groningen. Wouter J. Kiekens wants to thank Joseph Clements for proofreading the

manuscript. Chaïm la Roi acknowledges funding from two research projects funded by the Swedish Research Council for Health, Working Life and Welfare (FORTE) when preparing this article for submission: *YOUNG* [2012-1741] and *Interlocking Inequalities* [2016-07099].

Correspondence concerning this article should be addressed to Wouter J. Kiekens, Department of Sociology, Interuniversity Center for Social Science Theory and Methodology (ICS), University of Groningen, Grote Rozenstraat 31, 9712 TG, Groningen, the Netherlands. E-mail: w.j.kiekens@rug.nl

fined by observable physical characteristics or putative ancestry (Foner, 2018). Having a racial/ethnic minority status is especially relevant because it could make LGB people of color more susceptible as well as more resilient to sexual identity-related minority stressors affecting their mental health. On the one hand, sexual minority people of color might experience higher levels of minority stress, which may stem from strong traditional gender roles, conservative religious beliefs, or abandoning certain cultural values within their racial/ethnic community (Pachankis & Goldfried, 2004), and are more affected by these minority stressors compared to White sexual minority people (Moradi et al., 2010). On the other hand, people of color learn skills during their youth from their parents for coping with race/ethnicity-related stigma (Hughes et al., 2006), or acquire coping skills from experiencing racial/ethnic stigma-related stress themselves (Vaughn, Roesch, & Aldridge, 2009). These coping skills could help them to cope with sexual identity-related stressors later in life, whereas White LGB people do not acquire these skills.

Past research on racial/ethnic heterogeneity in sexual identity-based mental health disparities tended to use limited race/ethnicity categories (e.g., using a dichotomous racial/ethnic minority variable), often used U.S. adult samples, and often used samples that are not nationally representative. Therefore, we used three separate nationally representative data sets to examine differences in mental health (defined by psychological stress or strain) by sexual identity and racial/ethnic minority status in samples of U.S. adults, U.K. adults, and U.S. adolescents. Using these data sets provided us with the unique opportunity to explore the effects of the intersection of sexual identity and race/ethnicity across different national contexts and life phases and shed light on mental health differences in this at-risk group.

Mental Health of LGB People

Compared to heterosexual people, LGB people report poorer mental health (Plöderl & Tremblay, 2015). The minority stress framework is often used to explain these differences (Meyer, 2003). In the minority stress framework, four stigma-related stressors negatively affect LGB people's mental health. First, there are external/objective stressful events, which are explicit sources of stress, such as being harassed because of one's sexual identity. Vigilance to external/objective stressful events comprises a second type of minority stressor. Here, the stress stems from expecting and avoiding situations in which external/objective stressful events might happen. Internalized homophobia, which refers to negative attitudes in society against LGB people that are being directed to the self, forms the third type of minority stressor. Lastly, actively concealing one's sexual identity can be considered the fourth type of minority stressor, where stress stems from, for example, not being able to express one's identity and live an authentic life.

The psychological mediation framework (Hatzenbuehler, 2009), which extends the minority stress framework, is another framework explaining disparities in mental health between LGB and heterosexual people. The psychological mediation framework posits that LGB people are exposed to increased stress resulting from sexual identity-related stigma. These minority stressors create elevations in emotion dysregulation, social/interpersonal problems, and cognitive processes, resulting in higher risks for psychopathology. Thus, intra- and interpersonal psychological processes

mediate the relation between minority stressors and mental health (e.g., Kiekens et al., 2020; Timmins, Rimes, & Rahman, 2020).

There is ample evidence of stigma-related processes explaining the worse mental health of LGB adults. Research finds higher suicidal ideation in LGB adults who experience more homophobia and general violence and report a lower degree of outness (Plöderl et al., 2014). Furthermore, stigma-related stress increases the risk of depression (Timmins et al., 2020) and anxiety (Reitzel, Smith, Obasi, Forney, & Leventhal, 2017) in LGB adults. In community samples of LGB adults, gay-related stress (Lewis, Derlega, Griffin, & Krowinski, 2003) and discrimination (Fingerhut, Peplau, & Gable, 2010) were related to worse psychological well-being.

Not only for adults, but also among LGB adolescents, evidence of stigma-related processes explaining worse mental health is found. For example, higher rates of sexual identity-related victimization were associated with more suicidal ideation (Baams, Grossman, & Russell, 2015). Sexual identity-related stressful life events and negative attitudes toward homosexuality were related to anxious symptoms six months later among sexual minority youth (Rosario, Schrimshaw, Hunter, & Gwadz, 2002). Furthermore, stigma-related stress and sexual identity-related victimization increase the risk of depression in LGB adolescents as well (Mustanski, Andrews, & Puckett, 2016; Vanden Berghe, Dewaele, Cox, & Vincke, 2010). In schools, homophobic victimization by peers predicts mental health problems in LGB students, especially for girls (Poteat & Espelage, 2007).

Together, LGB people have poorer mental health than heterosexual people, and there is ample evidence to suggest that this disparity is a result of LGB people facing stigma-related stressors that their heterosexual counterparts do not experience. In summary, we expect that

H1: LGB people have poorer mental health than heterosexual people.

Intersecting Identities

Although research has found that LGB people have worse mental health compared to heterosexual people, some LGB people fair better than others. One factor that can attribute to this heterogeneity is having a racial/ethnic minority identity. From an intersectionality point of view, the experiences of people at the intersection of different social demographic categories, such as sexual identity and race/ethnicity, are shaped differently by social power (Bauer & Scheim, 2019; Crenshaw, 1989). That is, people that live at the intersection of specific sexual and racial/ethnic identities constitute a unique intersectional space that comes with its unique power and privileges, or lack thereof (Bauer, 2014), which could affect mental health.

In general, racial/ethnic discrimination is related to poorer mental well-being (Benner et al., 2018; Paradies et al., 2015), but some racial/ethnic minority groups often report better mental health compared with White people in general population samples. For example, a meta-analysis found no differences in the lifetime prevalence of major depressive disorders between White and Latino people and only slightly higher prevalence of depressive symptoms among Latino people (Menselson, Rehkopf, & Kubzansky, 2008). Similarly, Black people have repeatedly been found to report better mental health than White people (Barnes, Keyes, &

Bates, 2013; Erving, Thomas, & Frazier, 2019). Interestingly, the better mental health of people of color was not found in Europe, where racial/ethnic minorities report more depressive symptoms, especially in Western and Northern Europe (Missinne & Bracke, 2012). Thus, in the general population, people of color may report better mental health compared to White people, but how do minority sexual and racial/ethnic identities intersect to affect mental health?

One way sexual and racial/ethnic identities intersect to affect mental health negatively is explained in the greater risk perspective (Moradi et al., 2010). Here it is argued that sexual minority people of color experience higher levels of minority stress and are more affected by these stressors than White sexual minority people. It has been suggested that this greater risk for sexual minority people of color may stem from, for example, strong traditional gender roles, conservative religious beliefs, or the view that sexual minority people of color abandon certain cultural values (Pachankis & Goldfried, 2004). However, these more severe experiences of minority stress do not always result in differences in mental health between White and sexual minority people of color (Velez, Watson, Cox, & Flores, 2017).

Racial/ethnic minority identities might also intersect in such a way to make sexual minority people of color more resilient to minority stressors (Moradi et al., 2010). One way this manifests is by people of color being taught skills by their parents during their youth for coping with race/ethnicity-related stigma, referred to as ethnic socialization (Hughes et al., 2006). Parents can promote distrust, emphasizing the need for wariness in interracial interactions, which is communicated in cautions about barriers to success due to their race/ethnicity. Furthermore, parents make their children aware of discrimination and prepare them for coping with it, referred to as preparation for bias. These learned coping skills can help LGB people of color to cope with sexual identity-related stigma (Herek & Garnets, 2007). White LGB people are not racially socialized to cope with race/ethnicity-related stigma, making them less resilient to stigma. As such, the mental health of LGB people of color might suffer less from sexual identity-related stressors than the mental health of White LGB people.

Similarly, following the stress-related growth literature (Caplan, 1964), it is expected that LGB people of color learn skills from experience, enabling them to be able to cope with stigma-related stressors better than White LGB people. People can perceive certain stressors as learning opportunities with possible positive outcomes, such as acquiring coping skills (Cox, Dewaele, van Houtte, & Vincke, 2010). People of color can acquire specific coping skills from experiencing race/ethnicity-related stressors (Vaughn et al., 2009) that help them cope with sexual identity-related stressors later in life. White LGB people do not acquire these coping skills related to stigma early in life, which could result in them being more strongly affected by sexual identity-related stress.

Following expectancies proposed by the ethnic socialization and the stress-related growth literature, it is expected that the mental health of LGB people of color suffers less from sexual identity-related stressors than the mental health of White LGB people. Prior empirical research found that White LGB adults and adolescents report poorer mental health compared to LGB adults and adolescents of color (Bostwick et al., 2014; Consolacion, Russell, & Sue, 2004; Kertzner, Meyer, Frost, & Stirratt, 2009; Rodriguez-Seijas,

Eaton, & Pachankis, 2019; Russell & Truon, 2001), although some did not (Almazan, 2019). Taken together, we hypothesize the following:

H2: The negative association between an LGB identity and mental health is weaker among people of color than among White people.

The Present Study

We identified four limitations of research that examined how the intersection between sexual identity and race/ethnicity is associated with mental health. First, some research has used binary variables for race/ethnicity (e.g., Almazan, 2019; Kertzner et al., 2009; Velez et al., 2017), limiting our understanding of potential differences between ethnic/racial groups. Second, studies often use adult samples (e.g., Almazan, 2019; Kertzner et al., 2009; Rodriguez-Seijas et al., 2019; Velez et al., 2017), inhibiting our knowledge on the association between the intersection of sexual identity and race/ethnicity and mental health among adolescents. Such research among adolescents is especially relevant considering that sexual identity and race/ethnicity are (among others) characteristics of the self that are being shaped during adolescence, which is especially challenging as both identities might be stigmatized (Consolacion et al., 2004). Third, only four studies were identified that used nationally representative data sets (i.e., data sets representative of the population of a country; Almazan, 2019; Consolacion et al., 2004; Rodriguez-Seijas et al., 2019; Russell & Truon, 2001). Not using representative data sets can result in potential bias (Kuyper, Fernee, & Keuzenkamp, 2016). Last, research on this intersection and the association with mental health has primarily been conducted in the U.S., which has two drawbacks. First, although some studies use representative data, their findings can only be generalized to the U.S. Second, despite certain racial/ethnic groups facing similar barriers in both countries (Foner, 2018), different historical and social contexts have led to more political strategies, notions of rights, and public discourse on diversity in the U.S. compared to the U.K. (Better, 2008; Foner, 2018; Vertovec, 2007). This could lead to differences in how the intersection between sexual identity and race/ethnicity is associated with mental health between the U.K. and the U.S.

In order to overcome these limitations, we investigated the intersection of sexual identity and race/ethnicity, and its association with mental health in three nationally representative samples. The first data set is the 2011–2012 Understanding Society study, a longitudinal social survey from the U.K. The second is the 2015 National Survey on Drug Use and Health (NSDUH), a repeated cross-sectional study from the U.S. The third is the 2015 national Youth Risk Behavior Survey (YRBS), a repeated cross-sectional American study of adolescents in Grades 9 to 12. The benefit of analyzing national representative data sets of U.K. and U.S. adults is that it enabled us to present side-by-side assessments of the association between sexual identity and racial/ethnic minority identity, and mental health among adults in different contexts. Additionally, using a nationally representative data set of U.S. adolescents, we were also able to present side-by-side assessments of the association between sexual identity and race/ethnicity, and mental health in different life phases in the U.S.

Method

Data and Sample

U.K. adults. The third wave of the Understanding Society survey (N = 49,739) was used, which was collected in 2011–2012 (Lynn, 2009). This sample consists of the General Population Sample (GPS), the Ethnic Minority Boost Sample (EMBS), and the General Population Comparison (GPC). We only describe the data collection of the GPS because of space limitations and because the GPS comprised the largest part of the data and because information on the collection of the EMBS and the GPC is reported elsewhere (Lynn, 2009). The GPS for England, Scotland, and Wales was collected in two stages. In the first stage, postcode sectors were selected as the primary sampling units. Sectors with fewer than 500 residential addresses were grouped with an adjacent sector and treated as one postcode sector after that. The list of sectors was then sorted into 12 geographical strata, which were further subdivided based on the proportion of household reference persons classified as a nonmanual worker, the population density of the sector, and racial/ethnic minority density of the sector. In the second stage, a systematic random sample of 2,640 sectors was drawn from this sorted list. For 24 months, every month, 110 of these sectors were sampled. Within each of these 110 sectors, 18 addresses were selected to take part in the survey. For the Northern Ireland sample, a one-stage design was chosen where 2,395 addresses were selected from all domestic addresses. Only respondents older than 17 and who were personally interviewed were included in the present study, resulting in a final sample of N = $43,904 (M_{\text{age}} = 48.71, SD = 17.78).$

U.S. adults. The 2015 NSDUH study (N = 57,146) was selected with a five-stage sample design to create a sample of the civilian noninstitutionalized (e.g., people who are not imprisoned) population of the U.S. (Center for Behavioral Health Statistics & Quality, 2018). For the first stage, states were divided into state sampling regions (SSRs) based on composite size measures. Within each SSR, eight census tracts were selected, which served as the primary sampling unit. For the second stage, one census block group per census tract was selected with probability proportionate to a composite size measure. For the third sampling stage, each census block group was partitioned into small geographic areas composed of adjacent census blocks. These small geographic areas, or segments, are the tertiary sampling unit. One segment was selected within each census block group, with the probability proportionate to the segment's size. For the fourth sampling stage, dwelling units within each segment were selected to meet the minimum sample sizes for all age groups needed for the study. In the last stage, individuals within the dwelling units were selected. Interviewers visited each sampled dwelling unit to list the eligible individuals for this study. Only respondents older than 17 were included in the present study, resulting in a final sample of N =43,313 (*Mdn* age group = 26-34 [only age ranges were available in this dataset]).

U.S. adolescents. For the 2015 national YRBS (N = 15,625), a three-stage cluster sample design was used to create a sample of all public and private schools with students in Grades 9 to 12 in 50 states of the U.S. and the District of Columbia (Kann et al., 2016). The first stage consisted of 1,259 primary sampling units, which were counties, subareas of large counties, or groups of smaller

adjacent counties. These units were categorized into 16 strata according to their metropolitan statistical area status and the percentages of Black and Hispanic students in the sampling unit. Fifty-four of the 1,259 primary sampling units were then sampled with the probability of being sampled proportional to the overall school enrollment size for the primary sampling unit. In the second stage, from these 54 sampling units, 180 schools with students in any of the Grades 9-12 were sampled with the probability of being sampled proportional to school enrollment size. In the last stage, a random sample of one or two classrooms in each of Grades 9 to 12 was drawn within each participating school. The final sample consisted of N=15,122 respondents (Mdn age 16 [only age groups were available in this dataset]).

Measures

Independent variable: Sexual identity.

U.K. adults. Respondents were asked, "Which of the following options best describes how you think of yourself?" Response options were *heterosexual* (1), gay or lesbian (2), bisexual (3), other (4), and prefer not to say. Prefer not to say was coded as missing (n = 1,243). To facilitate better comparisons across data sets, respondents who answered other were removed from the data set (n = 406).

U.S. adults. Respondents' sexual identity was assessed by asking, "Which one of the following do you consider yourself to be?" Response options included *heterosexual (straight)* (1), *lesbian or gay* (2), *bisexual* (3), or *do not know* (4). Respondents who answered *do not know* were removed from the data set for better comparison across data sets (n = 248).

U.S. adolescents. Respondents were asked, "Which of the following best describes you?" Response options included *heterosexual (straight)* (1), *gay or lesbian* (2), *bisexual* (3), and *not sure* (4). Respondents who answered *not sure* were removed from the data set to improve comparability across data sets (n = 503).

Independent variable: Race/ethnicity.

U.K. adults. Respondents were asked to indicate their race/ethnicity. There were 17 response options (White: British, Irish, Gypsy/Traveler, other White; Asian: Indian, Pakistani, Bangladeshi, Chinese, other Asian; Black: Caribbean, African, other Black; Arab; Mixed: White and Caribbean, White and African, White and Asian; and other ethnic groups) of which respondents could choose one. These were recoded into four different race/ethnic groups: White (British, Irish, other White) (1); Black (Caribbean, African, other Black) (2); Asian (Indian, Pakistani, Bangladeshi, Chinese, other Asian) (3); and Multiracial/Other (Gypsy/Traveler, Arab, Mixed [White and Caribbean, White and African, White and Asian] and other ethnic group) (4).

U.S. adults. Race/ethnicity was measured by two questions. First, respondents were asked, "Are you of Hispanic, Latino, or Spanish origin or descent?" (yes/no). Next, respondents were asked, "Which of these groups describes you? Just give me the number or numbers from the card," with response categories White; Black or African American; American Indian or Alaska Native; Native Hawaiian, Guamanian or Chamorro, Samoan, other Pacific Islander, Asian, or other. Respondents could indicate multiple groups. This was recoded into White (1); Native (American Indian, Alaskan Native, Native Hawaiian, Guamanian or Chamorro, Samoan, and other Pacific Islander)

(2); Asian (3); Black (4); Hispanic (Hispanic and Hispanic with multiple groups) (5); and Multiracial (non-Hispanic and multiple groups) (6).

U.S. adolescents. Race/ethnicity was also measured with two questions for U.S. adolescents. First, respondents were asked, "Are you Hispanic or Latino?" (yes/no). Next, they were asked, "What is your race? (Select one or more responses)," with response categories American Indian or Alaska Native, Asian, Black or African American, Native Hawaiian or Other Pacific Islander, and White. A respondent was coded as Hispanic if they answered "yes" on the first question and did not respond to the second question. They were coded as Multiracial if they answered "no" on the first question and provided two or more answers on the second question. This yielded the following response categories, similar to the U.S. adults sample: White (1), Native (American Indian/Alaskan Native and Native Hawaiian and other Pacific Islander) (2), Asian (3), Black (4), Hispanic (Hispanic and Mixed Hispanic) (5), and Multiracial (Mixed non-Hispanic) (6).

Dependent variable: Mental health.

U.K. adults. Mental health was measured with the General Health Questionnaire (GHQ-12; Goldberg, 1978). Participants' answered 12 questions on mental health problems (e.g., "Have you recently felt constantly under strain?") using a response scale ranging from 1 (*better than usual*) to 4 (*much less than usual*). Responses were first recoded so that 0 was *better than usual* and 3 was *much less than usual*, then recoded so that higher scores reflected lower distress, and then responses were summed to derive a scale score ranging from 0 to 36 (Cronbach's alpha = 0.90). The GHQ-12 has been validated in the U.K., among other countries, as a measure to detect psychiatric disorders (McCabe, Thomas, Brazier, & Coleman, 1996; Shevlin & Adamson, 2005).

U.S. adults. Mental health was measured with the Kessler-6 (K6) distress scale (Kessler et al., 2002). Participants indicated how often they experienced symptoms of distress in the past 30 days (e.g., "...how often did you feel so sad or depressed that nothing could cheer you up?") using a rating scale ranging from 1 (*all of the time*) to 5 (*none of the time*). Responses were recoded so that 0 (*all of the time*) to 4 (*none of the time*) and responses were summed to create a scale score ranging from 0 to 24 so that higher scores reflected lower distress (Cronbach's alpha = 0.81). The K6 has been validated in the U.S., among other countries, as a measure to detect nonspecific psychological distress (Kessler et al., 2002).

U.S. adolescents. Mental health was measured by the item "During the past 12 months, did you ever feel so sad or hopeless almost every day for two weeks or more in a row that you stopped doing some usual activities?" (Brener et al., 2013). Response categories were *no* (0) and *yes* (1). This item approximates the Patient Health Questionnaire-2 (PHQ-2), which has been validated in the U.S. (McCabe et al., 1996).

Covariates. When studying mental health, it is important to take into account certain demographic variables such as gender, age, education, and income as research indicates that these can contribute to differences in mental health (Allen, Balfour, Bell, & Marmot, 2014; Gadalla, 2009). For both U.K. and U.S. adults, gender, age, level of education, and income were therefore used as covariates. For U.S. adolescents, only gender and age were used as covariates because education and income were not measured for both adolescents and their parents.

Analyses

Covariates were controlled for in all analyses to obtain unbiased estimates. Design variables and weights were used to correct in all analyses for the effects of the sampling design. Linear regression analyses were performed for U.K. and U.S. adults, providing an informal comparison of how sexual identity and race/ethnicity are associated with average mental health among U.K. and U.S. adults. In order to improve comparability, mental health was standardized to have M = 0 and SD = 1 in both samples. Data for U.S. adolescents were analyzed using logistic regression analyses. When groups are compared in logistic regression analysis, different log-odds or odds ratios between groups can reflect differences in effect, but also differences in unobserved heterogeneity (Mood, 2010). Therefore, predicted probabilities were estimated in order to compare the probability of impaired mental health for different sexual identity and race/ethnicity groups. The predicted probabilities were estimated by calculating marginal effects, keeping the covariates at their mean level. Results held when the effects of the covariates were set to their minimum and maximum values.

Due to small cell size, it was not possible to conduct the analyses with categorical variables for both sexual identity and race/ethnicity. Dichotomizing both variables would obstruct our understanding of how sexual identity and race/ethnicity intersect to affect mental health, thereby erasing subgroup differences. Such dichotomization of categorical identity variables is also referred to as the lumping error (Else-Quest & Hyde, 2016). Therefore, to avoid this error but maximize comparison of categories, all analyses were conducted in two ways: (1) with a categorical operationalization of sexual identity (1 = Heterosexual, 2 = Lesbian/ gay, and 3 = Bisexual) and a dichotomous race/ethnicity variable (0 = White and 1 = Person of color) (presented in Model 1 in Tables 2, 3, and 4), and (2) with a dichotomous sexual identity variable (0 = Heterosexual and 1 = Sexual minority) and a categorical race/ethnicity operationalization (for U.K. adults, 1 = White, 2 = Black, 3 = Asian, and 4 = Multiracial; for U.S. adults and adolescents, 1 = White, 2 = Native, 3 = Asian, 4 = Black, 5 = Hispanic, and 6 = Multiracial) (presented in Model 2 in Tables 2, 3, and 4).

Missing Data

Missing data analyses on all three samples suggested that missing data were missing at random. Multiple imputation is a sufficient procedure to take into account this type of missingness (Schafer & Graham, 2002). Furthermore, it could be the case that some of the respondents who replied *prefer not to say* on the sexual identity question in the U.K. adults sample did so in order to hide a stigmatized LGB sexual identity. This would have led to a correlation between sexual identity and missing sexual identity information in that dataset (i.e., "Missing Not at Random"). However, if this were the case, multiple imputation would still be the optimal method for dealing with this missingness (e.g., Pedersen et al., 2017). Twenty-five imputed data sets were created for each sample, which is likely sufficient given the modest proportion of missing data across data sets (see Table 1; Liu & De, 2015).

This document is copyrighted by the American Psychological Association or one of its allied publishers. This article is intended solely for the personal use of the individual user and is not to be disseminated broadly.

Table 1

Descriptive Statistics by Sample

	U.K.	ıdults (A	U.K. adults $(N = 43,904)$	4)	U.S. adults $(N = 43,313)$	V = 43,	313)		U.S. adolescents $(N = 15,122)$	nts $(N = 1)$	5,122)	
Variable	M	QS	Range	n (% missing)	M	CS	Range	n (% missing)	M	SD Range	1 1	n (% missing)
Mental health	24.91	5.50	0–36	38,731 (11.8%)	19.60	4.71	0-24	43,313 (0.0%)	Yes No	30.5	14,	14,966 (1.0%)
Sexual identity	Heterosexual Lesbian/Gay Bisexual	97.8		37,485 (14.6%)	Heterosexual Lesbian/Gay Bisexual	93.8		42,802 (1.2%)	erosexual bian/Gay	91.2 2.3 6.5	14,	14,200 (6.1%)
Race/ethnicity	White White Black Asian Multiracial	86.1 3.7 8.0 2.2		43,648 (0.6%)	Discourant White Native Asian Black	59.9 2.0 4.6		43,313 (0.0%)		45.0 1.7 4.1	14,	14,784 (2.2%)
					Hispanic Multiracial	3.3			nic acial	33.4		
Gender	Men Women	43.9		43,904 (0.0%)	Men Women	45.6 54.4		43,313 (0.0%)		49.7 50.3	15,	15,011 (0.7%)
Age		17.78	18–103 43,997	43,997 (0.0%)	18–25 years old 26–34 years old 35–49 years old 50–64 years old 65 or older	33.4 20.9 25.6 11.8 8.2		43,313 (0.0%)	s old or younger old	24.5 26.0 24.7 13.6	14,	14,055 (0.4%)
Education	No qualification Other qualification GCSE A level Other degree Degree	14.9 10.0 20.1 20.5 11.7 22.8		43,784 (0.3%)	No high school diploma High school diploma/GED Some college credit, no degree Associates degree College graduate or higher	14.3 27.0 24.1 9.3 25.3		43,289 (0.1%)				
Income	£0-1,000 £1,000-2,000 £2,000-3,000 £3,000-4,000 £4,000 or more	37.3 34.8 15.8 6.3 5.7		43,904 (0.0%)	Less than \$10,000 \$10,000 till \$19,999 \$20,000 till \$29,999 \$30,000 till \$39,999 \$40,000 till \$74,999 \$75,000 or more	31.4 20.1 12.8 9.4 7.7 9.5		41,796 (3.5%)				

Note. For categorical variables, percentages are given.

Table 2
Linear Regression of Mental Health by Sexual Identity and Race/Ethnicity Among U.K. Adults

b	95% CI	1.	
		b	95% CI
0.39***	[-0.44, -0.33]	-0.39***	[-0.44, -0.34]
		-0.44***	[-0.55, -0.32]
0.33***	[-0.46, -0.19]		
0.59***	[-0.77, -0.41]		
0.01	[-0.05, 0.03]		
		0.04	[-0.03, 0.11]
		-0.02	[-0.08, 0.03]
		-0.05	[-0.13, 0.04]
0.05	[-0.35, 0.45]		
0.22	[-0.17, 0.61]		
	. , ,		
		-0.26	[-0.69, 0.18]
		0.29	[-0.09, 0.66]
		0.17	[-0.38, 0.73]
0.00***	[0.00, 0.00]	0.00***	
0.03***	[0.02, 0.04]		[0.02, 0.04]
0.04***	[0.03, 0.05]	0.04***	[0.03, 0.05]
	0.33*** 0.59*** 0.01 0.05 0.22 0.00*** 0.19***	0.33*** [-0.46, -0.19] 0.59*** [-0.77, -0.41] 0.01 [-0.05, 0.03] 0.05 [-0.35, 0.45] 0.22 [-0.17, 0.61] 0.00*** [0.00, 0.00] 0.19*** [0.17, 0.22] 0.03*** [0.02, 0.04]	0.33*** [-0.46, -0.19] 0.59*** [-0.77, -0.41] 0.01 [-0.05, 0.03] 0.04 -0.02 -0.05 0.05 [-0.35, 0.45] 0.22 [-0.17, 0.61] -0.26 0.29 0.17 0.00*** [0.00, 0.00] 0.19*** [0.17, 0.22] 0.03*** [0.02, 0.04]

Note. Model 1 uses a categorical sexual identity variable and a dichotomous race ethnicity variable. Model 2 uses a dichotomous sexual identity variable and a categorical race ethnicity variable. CI = confidence interval. $^*p < 0.05$. $^{**}p < 0.01$. $^{***}p < 0.001$.

Results

Descriptive Statistics

The descriptive statistics of all three samples are reported in Table 1.

U.K. adults. Mental health was skewed to the left, indicating that most respondents had good mental health. Concerning sexual identity, 2.2% of the respondents self-identified as LGB, which is similar to other studies using population-based samples (Gates, 2011). A vast majority of the respondents (86.1%) identified as White.

U.S. adults. Mental health was skewed to the left among U.S. adults too. A total of 6.2% of the respondents identified as LGB, which is somewhat higher than other population estimations (Gates, 2011). Within this sample, 59.9% of the respondents identified as White, making the sample sufficiently racially/ethnically diverse.

U.S. adolescents. In total, 35.0% of the respondents indicated having low mental health. Of all respondents, 8.8% identified as LGB, which is higher than in previous population-based samples (Gates, 2011), although these population-based samples consisted of adults. The sample was racially/ethnically diverse, with 45.0% of respondents identifying as White.

Regression Analyses

U.K. adults linear regression. Focusing on Model 1 in Table 2, lesbian/gay (b = -0.33, 95% CI [-0.46, -0.19]) and bisexual people (b = -0.59, 95% CI [-0.77, -0.41]) reported poorer

mental health compared to heterosexual people, which supports the first hypothesis. As dependent variables were standardized, this means that lesbian/gay people reported -0.33~SD and bisexual people reported -0.59~SD lower mental health compared to heterosexual respondents, indicating that the effect for bisexual people is more pronounced. No significant interaction effects with race/ethnicity were found in Models 1 and 2, finding no support for the second hypothesis. Furthermore, we found positive associations of age (b=0.00, 95% CI [0.00, 0.00]), gender (b=0.19, 95% CI [0.17, 0.22]), education (b=0.03, 95% CI [0.02, 0.04]), and income (b=0.04, 95% CI [0.03, 0.05]) with mental health (see Model 1).

U.S. adults linear regression. As expected in hypothesis 1, lesbian/gay (b = -0.28, 95% CI [-0.43, -0.14]) and bisexual people (b = -0.61, 95% CI [-0.71, -0.50]) had significantly worse mental health compared to heterosexual people (see Model 1, Table 3). Thus, lesbian/gay people reported -0.28 SDand bisexual people reported -0.61 SD lower mental health compared to heterosexual respondents, indicating that the effect for bisexual people is more pronounced. There was no significant interaction effect between the categorical sexual identity variable and the dichotomous race/ethnicity variable in Model 1, nor was there a significant interaction between the dichotomous sexual identity variable and the categorical race/ethnicity variable in Model 2. Thus, there was no support for the second hypothesis. Furthermore, we found positive associations of age (b = 0.13, 95% CI [0.12, 0.14]), gender (b = 0.08, 95% CI)[0.05, 0.11]), and income (b = 0.06, 95% CI [0.05, 0.07]) with mental health (see Model 1).

Table 3
Linear Regression of Mental Health by Sexual Identity and Race/Ethnicity Among U.S. Adults

	1	Model 1	1	Model 2
Variable	b	95% CI	b	95% CI
Intercept	-0.71***	[-0.76, -0.65]	-0.75***	[-0.81, -0.68]
Sexual identity binary (Heterosexual = reference) Sexual minority			-0.46***	[-0.55, -0.37]
Sexual identity categorical (Heterosexual = reference) Lesbian/gay Bisexual		[-0.43, -0.14] [-0.71, -0.50]		
Race/ethnicity binary (White = reference) Person of color		[0.11, 0.17]		
Race/ethnicity categorical (White = reference) Native Asian Black Hispanic Multiracial Categorical sexual identity × Binary race/ethnicity Lesbian/gay × Person of color Bisexual × Person of color Binary sexual identity × Categorical race/ethnicity	-0.04	[-0.27, 0.19] [-0.06, 0.29]	0.00 0.09** 0.10*** 0.23*** -0.15**	
Sexual minority × Native Sexual minority × Asian Sexual minority × Black Sexual minority × Hispanic Sexual minority × Multiracial	0.12***	F0.10. 0.147	0.05 0.11 0.14 -0.05 0.06	$ \begin{bmatrix} -0.41, 0.52 \\ -0.13, 0.35 \end{bmatrix} $ $ \begin{bmatrix} -0.03, 0.31 \\ -0.24, 0.14 \end{bmatrix} $ $ \begin{bmatrix} -0.23, 0.36 \end{bmatrix} $
Age Gender	0.13*** 0.08***	[0.12, 0.14] [0.05, 0.11]	0.13*** 0.08***	[0.12, 0.14]
Education	0.08	[-0.00, 0.11]	0.08	[0.06, 0.11] $[-0.00, 0.01]$
Income	0.06***		0.06***	[0.05, 0.07]

Note. Model 1 uses a categorical sexual identity variable and a dichotomous race ethnicity variable. Model 2 uses a dichotomous sexual identity variable and a categorical race ethnicity variable. CI = confidence interval. p < 0.05. *** p < 0.01. **** p < 0.001.

U.S. adolescents logistic regression. Following the first hypothesis, lesbian/gay (p = .47, 95% CI [.37, .57]) and bisexual adolescents (p = .59, 95% CI [.53, .65]) were significantly more likely than heterosexual adolescents (p = .26, 95% CI [.25, .28]) to report impaired mental health (Table 4, Model 1). Again, this effect was most pronounced for bisexual adolescents. Significant interaction effects between the categorical sexual identity variable and the dichotomous race/ethnicity variable were found in Model 1. That is, lesbian/gay adolescents of color (p = .38, 95% CI [.25, .51]) were less likely than White lesbian/gay adolescents (p = .57, 95% CI [.43, .70]) to report impaired mental health, which is in line with the second hypothesis. Similarly, bisexual adolescents of color (p = .54, 95% CI [.46, .61]) were less likely than White bisexual adolescents (p = .64, 95% CI [.56, .72]) to report impaired mental health. When the dichotomous sexual identity variable and the categorical race/ethnicity variable were used (see Model 2), it was found that Black sexual minority adolescents (p =.40, 95% CI [.29, .51]) were less likely than White sexual minority adolescents (p = .63, 95% CI [.56, .71]) to report impaired mental health, in support of the second hypothesis. No other significant differences in impaired mental health were found between White sexual minority adolescents and sexual minority adolescents of color. Of note, adolescents of color (p = .39, 95% CI [.34, .45])were less likely to report impaired mental health compared to White adolescents (p = .48, 95% CI [.41, .54]) (See Model 1). When a categorical race/ethnicity variable was used, only Black (p=.31, 95% CI [.25, .37]) adolescents were less likely to report impaired mental health than White adolescents (p=.42, 95% CI [.38, .47]; see Model 2). Lastly, heterosexual Hispanic (p=.32, 95% CI [.19, .35]) and Multiracial (p=.34, 95% CI [.30, .39]) adolescents were more likely than White heterosexual adolescents (p=.24, 95% CI [.21, .27]) to report impaired mental health (See Model 2).

Discussion

This study aimed to investigate how the association between sexual identity and mental health differs based on race/ethnicity in national representative samples of U.K. adults, U.S. adults, and U.S. adolescents. Following the minority stress (Meyer, 2003) and psychological mediation framework (Hatzenbuehler, 2009), we expected LGB people to report worse mental health than heterosexual people. Furthermore, we hypothesized that the negative association between an LGB identity and mental health is weaker among people of color than among White people. The benefit of analyzing national representative data sets of U.K. adults, U.S. adults and U.S. adolescents is that this enabled us to present side-by-side assessments of the association between sexual identity and race/ethnicity, and mental health among adults in different contexts, as well as in different life phases in the U.S. simultaneously.

Table 4

Predicted Probabilities of Impaired Mental Health by Sexual Identity and Race/Ethnicity Among
U.S. Adolescents

	M	odel 1	M	odel 2
Variable	p	95% CI	p	95% CI
Sexual identity binary ^a				
Heterosexual			.28	[.26, .31]
Sexual minority			.56***	[.46, .66]
Sexual identity categorical ^a				
Heterosexual	.26	[.25, .28]		
Lesbian/gay	.47***	[.37, .57]		
Bisexual	.59***	[.53, .65]		
Race/ethnicity binary ^b				
White	.48	[.41, .54]		
Person of color	.39*	[.34, .45]		
Race/ethnicity categorical ^b				
White			.42	[.38, .47]
Native			.57	[.35, .79]
Asian			.35	[.22, .47]
Black			.31***	[.25, .37]
Hispanic			.42	[.37, .47]
Multiracial			.42	[.33, .51]
Categorical sexual identity × Binary race/ethnicity ^c				
Heterosexual and White	.24	[.21, .27]		
Heterosexual and Person of color	.29**	[.26, .31]		
Lesbian/gay and White	.57	[.43, .70]		
Lesbian/gay and Person of color	.38*	[.25, .51]		
Bisexual and White	.64	[.56, .72]		
Bisexual and Person of color	.54*	[.46, .61]		
Binary sexual identity \times Categorical race/ethnicity ^d				
Heterosexual and White			.24	[.21, .27]
Sexual minority and White			.63	[.56, .71]
Heterosexual and Native			.36	[.24, .49]
Sexual minority and Native			.75	[.44, .99]
Heterosexual and Asian			.21	[.16, .27]
Sexual minority and Asian			.51	[.25, .76]
Heterosexual and Black			.23	[.20, .26]
Sexual minority and Black			.40***	[.29, .51]
Heterosexual and Hispanic			.32***	[.29, .35]
Sexual minority and Hispanic			.53	[.45, .61]
Heterosexual and Multiracial			.34***	[.30, .39]
Sexual minority and Multiracial			.50	[.34, .67]

Note. Model 1 uses a categorical sexual identity variable and a dichotomous race ethnicity variable. Model 2 uses a dichotomous sexual identity variable and a categorical race ethnicity variable. Age and gender were set at their mean values for these analyses. CI = confidence interval.

As hypothesized, we found that among U.K. adults, U.S. adults, and U.S. adolescents, LGB people reported worse mental health than heterosexual people. The poorest mental health was reported among bisexual people, in line with previous research (e.g., Plöderl & Tremblay, 2015). This might reflect the double stigmatization that bisexual people face from both the heterosexual and the LGB community, commonly referred to as biphobia (Hayfield, 2020). This study was uniquely able to show variation in sexual identity-based disparities in mental health by race/ethnicity for adults in the U.K. and U.S., as well as for adults and adolescents in the U.S., using national representative data.

We also hypothesized that the negative association between LGB identity and mental health is weaker among people of color than among White people. For U.K. and U.S. adults, we found no support for this hypothesis. In the logistic regression analyses for U.S. adolescents, some support was found for this hypothesis. That is, when a categorical sexual identity variable and a dichotomous race/ethnicity variable were used, we found that lesbian/gay and bisexual adolescents of color were less likely to report an impaired mental health than White lesbian/gay and bisexual adolescents (respectively). Using a categorical race/ethnicity variable showed that only Black sexual minority adolescents were less likely to report impaired mental health compared to White sexual minority adolescents. Thus, for LGB adolescents as a group and for Black U.S. adolescents specifically, support for the second hypothesis was found. That this hypothesis only held for Black sexual minor-

^a Reference for the significance test was the heterosexual group. ^b Reference for the significance test was the White group. ^c Within each sexual orientation group, the reference for the significance test was the White group. ^d Within each race/ethnicity group, the reference group for the significance test was White group of the same sexual identity.

^{*} p < .05. ** p < .01. *** p < .001.

ity adolescents might reflect differences in racial socialization among people of color in the U.S. For example, Black adolescents receive more preparation for bias messages compared to Asian or Latinx Americans (Priest et al., 2014). Future research may empirically test if such racial socialization or preparation for encountering racism explains the difference in mental health between LGB adolescents of color and White LGB adolescents.

Contrasting U.K. and U.S. Adults

Among U.K. and U.S. adults, having a lesbian/gay or bisexual identity was related to worse mental health. Furthermore, despite the different historical and social contexts of people of color in the U.K. and the U.S. (Better, 2008; Vertovec, 2007), no support was found for a weaker association between a LGB identity among people of color than among White LGB people in either country. The fact that similar findings were observed among adults in both countries might be related to age. That is, the health benefits of processes such as ethnic socialization (Hughes et al., 2006) and stress-related growth (Caplan, 1964) could be especially salient during adolescence. In other words, White LGB people could become more resilient to the negative effects of sexual identity-related stigma in adulthood, diminishing the benefits ethnic socialization and stress-related growth provide LGB people of color.

Contrasting U.S. Adolescents and Adults

There were some differences between the results of U.S. adults and adolescents. Among U.S. adults, the negative association between LGB identity and mental health was not weaker among people of color than among White people. Among U.S. adolescents, such an effect was found among lesbian/gay adolescents of color (compared to White lesbian/gay adolescents), among bisexual adolescents of color (compared to White bisexual adolescents), and among Black sexual minority adolescents (compared to White sexual minority adolescents). This aligns with the idea that the health benefits of ethnic socialization (Hughes et al., 2006) and stress-related growth (Caplan, 1964) are especially relevant in adolescence but do not last into adulthood. Studies using longitudinal data can investigate if such health benefits decrease in the transition to young adulthood. Furthermore, the better mental health among Black sexual minority adolescents contradicts expectancies from the greater risk perspective (Moradi et al., 2010). This could potentially indicate that ethnic socialization and stressrelated growth mechanisms more strongly affect mental health among Black sexual minority adolescents.

Limitations and Future Directions

One of the strengths of this study is that it analyzed the associations between sexual identity and the intersection with race/ethnicity using three nationally representative samples, enabling us to extrapolate our findings to the general population. The data, however, do not allow for tests of the mechanisms behind studied associations. Therefore, we invite future nationally representative studies to collect data that allow for tests of minority stress, ethnic socialization, and stress-related growth mechanisms.

A limitation of using three different data sets is the different operationalizations of both dependent and independent variables.

With regard to sexual identity, we tried to circumvent this by limiting our analyses to groups that were measured across all studies (i.e., heterosexual, lesbian/gay, and bisexual). This led to the exclusions of the "other," "don't know," and "not sure" categories and inhibited us from studying the previously found health disparities of people who are uncertain about their sexual identity (Bejakovich & Flett, 2018) and people with a different sexual minority identity such as a pansexual identity (Greaves, Sibley, Fraser, & Barlow, 2019). Race/ethnicity was assessed differently across studies as well. Similar categories could be created for the two U.S. samples but not for the U.K. sample. However, this was to be expected because of the different racial/ethnic composition in the U.K. and U.S. (Better, 2008; Vertovec, 2007). Lastly, the measures of mental health were different across all studies. Within the constraints of every data set, we tried to choose the best-suited measures of mental health. There are differences between these measures; for example, the K6 scale used in the U.S. adults sample is better able to detect mood and anxiety disorders than the GHQ-12 scale used in the U.K. adults sample (Furukawa, Kessler, Slade, & Andrews, 2003). Nevertheless, in order to harmonize the operationalization of dependent variables across data sets, the continuous mental health measures in the U.K. and U.S. adult samples were standardized. For U.S. adults and adolescents, different types of statistical analyses were performed, making it harder to contrast associations between these two data sets.

A different limitation of the present study is that because no U.K. adolescent data set measuring constructs of interest was available, only a dataset with U.K. adults was used. Hence, we were not able to study the intersection of sexual identity and race/ethnicity for adults and adolescents between and within two countries. Thus, it cannot be determined if the lack of between-country differences we found for adult respondents would be replicated among samples of adolescents. Furthermore, we do not know whether observed differences between adults and adolescents in the U.S. would correspond with differences between adolescents and adults in the U.K. Related to this issue, national representative data sets from two different countries were used, and therefore generalizing these findings to other countries may not be possible.

Although the current study set out to research the intersection of sexual identity and race/ethnicity on mental health, we were not able to conduct analyses with categorical operationalizations of both sexual identity and race/ethnicity due to small cell sizes. Not being able to dissect the effects of intersecting identities completely is a known drawback of doing intersectional research with quantitative data (Bostwick et al., 2014; Bowleg, 2012) and sometimes referred to as the lumping error (Else-Quest & Hyde, 2016). Data collections focusing on (over)sampling these minority groups could help to overcome this drawback. Future research should focus on further dissecting the effects of intersecting identities, moving beyond omnibus measures of sexual identity and race/ ethnicity. Additionally, we were restricted in our racial/ethnic group comparisons. For example, there was no Middle Eastern response category in both U.S. data sets, limiting the knowledge of these groups.

Furthermore, the lower probability of impaired mental health among LGB adolescents of color could also be a result of cultural relativity (Neighbors, Jackson, Campbell, & Williams, 1989), which assumes that there are cultural differences in how White

people and people of color express psychopathology. This could result in overreporting of mental health problems among White people, and underreporting among people of color. Thus, differences in mental health between White LGB people and LGB people of color could also result from differences in reporting psychopathology. This should be a factor future research can consider when studying the intersection of stigma and how it affects mental health.

Findings from this study have implications for mental health professionals who work with LGB people. They should be aware of the larger mental health disparity bisexual people report compared to heterosexual people. Furthermore, they should be aware that White LGB adolescents might lack certain skills to cope with stigma compared with LGB adolescents of color, making them less resilient. Although these effects were found in the U.S., mental health professionals outside of the U.S. should also be aware of how sexual identity and race/ethnicity might intersect in their association with mental health.

In conclusion, the current study uniquely showed that LGB people have a higher risk of impaired mental health compared to heterosexual people in both the U.K. and the U.S. Among U.S. adolescents, LGB adolescents of color reported better mental health compared with White LGB adolescents. More specifically, Black LGB adolescents reported better mental health compared to White LGB adolescents. Together, the present study contributes to a better understanding of the relation between intersecting identities and mental health.

References

- Allen, J., Balfour, R., Bell, R., & Marmot, M. (2014). Social determinants of mental health. *International Review of Psychiatry*, 26, 392–407. http://dx.doi.org/10.3109/09540261.2014.928270
- Almazan, E. (2019). Are Black sexual minority adults more likely to report higher levels of psychological distress than White sexual minority adults? Findings from the 2013–2017 National Health Interview Survey. *Social Sciences*, 8, 14. http://dx.doi.org/10.3390/socsci8010014
- Baams, L., Grossman, A. H., & Russell, S. T. (2015). Minority stress and mechanisms of risk for depression and suicidal ideation among lesbian, gay, and bisexual youth. *Developmental Psychology*, 51, 688–696. http://dx.doi.org/10.1037/a0038994
- Barnes, D. M., Keyes, K. M., & Bates, L. M. (2013). Racial differences in depression in the United States: How do subgroup analyses inform a paradox? Social Psychiatry and Psychiatric Epidemiology, 48, 1941– 1949. http://dx.doi.org/10.1007/s00127-013-0718-7
- Bauer, G. R. (2014). Incorporating intersectionality theory into population health research methodology: Challenges and the potential to advance health equity. *Social Science & Medicine*, 110, 10–17. http://dx.doi.org/10.1016/j.socscimed.2014.03.022
- Bauer, G. R., & Scheim, A. I. (2019). Methods for analytic intercategorical intersectionality in quantitative research: Discrimination as a mediator of health inequalities. *Social Science & Medicine*, 226, 236–245. http://dx.doi.org/10.1016/j.socscimed.2018.12.015
- Bejakovich, T., & Flett, R. (2018). "Are you sure?": Relations between sexual identity, certainty, disclosure, and psychological well-being. *Journal of Gay & Lesbian Mental Health*, 22, 139–161. http://dx.doi.org/10.1080/19359705.2018.1427647
- Benner, A. D., Wang, Y., Shen, Y., Boyle, A. E., Polk, R., & Cheng, Y. (2018). Racial/ethnic discrimination and well-being during adolescence: A meta-analytic review, 73, 855–883.
- Better, S. (2008). *Institutional racism*. Lanham, MD: Rowman & Little-field.

- Bostwick, W. B., Meyer, I., Aranda, F., Russell, S., Hughes, T., Birkett, M., & Mustanski, B. (2014). Mental health and suicidality among racially/ethnically diverse sexual minority youths. *American Journal of Public Health*, 104, 1129–1136. http://dx.doi.org/10.2105/AJPH.2013.301749
- Bowleg, L. (2012). The problem with the phrase women and minorities: Intersectionality-an important theoretical framework for public health. *American Journal of Public Health*, 102, 1267–1273. http://dx.doi.org/ 10.2105/AJPH.2012.300750
- Brener, N. D., Kann, L., Shanklin, S., Kinchen, S., Eaton, D. K., Hawkins, J., . . . the Centers for Disease Control and Prevention (CDC). (2013).
 Methodology of the Youth Risk Behavior Surveillance System—2013.
 MMWR Recommendations and Reports, 62, 1–20.
- Caplan, G. (1964). Principles of preventive psychiatry. New York, NY: Basic Books.
- Center for Behavioral Health Statistics and Quality. (2018). 2015 National Survey on Drug Use and Health Public Use File Codebook. Rockville, MD.
- Consolacion, T. B., Russell, S. T., & Sue, S. (2004). Sex, race/ethnicity, and romantic attractions: Multiple minority status adolescents and mental health. *Cultural Diversity and Ethnic Minority Psychology*, 10, 200–214. http://dx.doi.org/10.1037/1099-9809.10.3.200
- Cox, N., Dewaele, A., van Houtte, M., & Vincke, J. (2010). Stress-related growth, coming out, and internalized homonegativity in lesbian, gay, and bisexual youth. An examination of stress-related growth within the minority stress model. *Journal of Homosexuality*, 58, 117–137. http:// dx.doi.org/10.1080/00918369.2011.533631
- Crenshaw, K. (1989). Demarginalizing the intersection of race and sex: A Black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics. *University of Chicago Legal Forum*, 1989, 139– 168.
- Else-Quest, N. M., & Hyde, J. S. (2016). Intersectionality in quantitative psychological research: II. Methods and techniques. *Psychology of Women Quarterly*, 40, 319–336. http://dx.doi.org/10.1177/036 1684316647953
- Erving, C. L., Thomas, C. S., & Frazier, C. (2019). Is the Black–White mental health paradox consistent across gender and psychiatric disorders? *American Journal of Epidemiology*, 188, 314–322. http://dx.doi.org/10.1093/aje/kwy224
- Fingerhut, A. W., Peplau, L. A., & Gable, S. L. (2010). Identity, minority stress and psychological well-being among gay men and lesbians. *Psychology and Sexuality*, 1, 101–114. http://dx.doi.org/10.1080/19419899 .2010.484592
- Foner, N. (2018). Race in an era of mass migration: Black migrants in Europe and the United States. *Ethnic and Racial Studies*, 41, 1113–1130. http://dx.doi.org/10.1080/01419870.2018.1411600
- Furukawa, T. A., Kessler, R. C., Slade, T., & Andrews, G. (2003). The performance of the K6 and K10 screening scales for psychological distress in the Australian National Survey of Mental Health and Well-Being. *Psychological Medicine*, 33, 357–362. http://dx.doi.org/10.1017/ S0033291702006700
- Gadalla, T. M. (2009). Determinants, correlates and mediators of psychological distress: A longitudinal study. Social Science & Medicine, 68, 2199–2205. http://dx.doi.org/10.1016/j.socscimed.2009.03.040
- Gates, G. J. (2011). How many people are lesbian, gay, bisexual, and transgender? Los Angeles, CA: The Williams Institute.
- Goldberg, D. P. (1978). Manual for the General Health Questionnaire. Windsor, England: NFER-Nelson.
- Greaves, L. M., Sibley, C. G., Fraser, G., & Barlow, F. K. (2019). Comparing pansexual- and bisexual-identified participants on demographics, psychological well-being, and political ideology in a New Zealand national sample. *Journal of Sex Research*, 56, 1083–1090. http://dx.doi.org/10.1080/00224499.2019.1568376

- Hatzenbuehler, M. L. (2009). How does sexual minority stigma "get under the skin"? A psychological mediation framework. *Psychological Bulle*tin, 135, 707–730. http://dx.doi.org/10.1037/a0016441
- Hayfield, N. (2020). Bisexual and asexual identities: Exploring and challenging invisibility and invalidation. London, England: Routledge. http://dx.doi.org/10.4324/9780429464362
- Herek, G. M., & Garnets, L. D. (2007). Sexual orientation and mental health. *Annual Review of Clinical Psychology*, 3, 353–375. http://dx.doi.org/10.1146/annurev.clinpsy.3.022806.091510
- Hughes, D., Rodriguez, J., Smith, E. P., Johnson, D. J., Stevenson, H. C., & Spicer, P. (2006). Parents' ethnic-racial socialization practices: A review of research and directions for future study. *Developmental Psychology*, 42, 747–770. http://dx.doi.org/10.1037/0012-1649.42.5.747
- Kann, L., Olsen, E. O., McManus, T., Harris, W. A., Shanklin, S. L., Flint, K. H., . . . Zaza, S. (2016). Sexual identity, sex of sexual contacts, and health-related behaviors among students in grades 9–12—United States and selected sites, 2015. Atlanta, GA: Centers for Disease Control and Prevention
- Kertzner, R. M., Meyer, I. H., Frost, D. M., & Stirratt, M. J. (2009). Social and psychological well-being in lesbians, gay men, and bisexuals: The effects of race, gender, age, and sexual identity. *American Journal of Orthopsychiatry*, 79, 500–510. http://dx.doi.org/10.1037/a0016848
- Kessler, R. C., Andrews, G., Colpe, L. J., Hiripi, E., Mroczek, D. K., Normand, S.-L. T., . . . Zaslavsky, A. M. (2002). Short screening scales to monitor population prevalences and trends in non-specific psychological distress. *Psychological Medicine*, 32, 959–976. http://dx.doi.org/ 10.1017/S0033291702006074
- Kiekens, W., la Roi, C., Bos, H. M. W., Kretschmer, T., van Bergen, D. D., & Veenstra, R. (2020). Explaining health disparities between heterosexual and LGB adolescents by integrating the minority stress and psychological mediation frameworks: Findings from the TRAILS Study. *Jour*nal of Youth and Adolescence. Advance online publication. http://dx.doi .org/10.1007/s10964-020-01206-0
- King, M., Semlyen, J., Tai, S. S., Killaspy, H., Osborn, D., Popelyuk, D., & Nazareth, I. (2008). A systematic review of mental disorder, suicide, and deliberate self harm in lesbian, gay and bisexual people. *BMC Psychiatry*, 8, 70. http://dx.doi.org/10.1186/1471-244X-8-70
- Kuyper, L., Fernee, H., & Keuzenkamp, S. (2016). A comparative analysis of a community and general sample of lesbian, gay, and bisexual individuals. *Archives of Sexual Behavior*, 45, 683–693. http://dx.doi.org/ 10.1007/s10508-014-0457-1
- Lewis, R. J., Derlega, V. J., Griffin, J. L., & Krowinski, A. C. (2003). Stressors for gay men and lesbians: Life stress, gay-related stress stigma consciousness, and depressive symptoms. *Journal of Social and Clinical Psychology*, 22, 716–729. http://dx.doi.org/10.1521/jscp.22.6.716 .22932
- Liu, Y., & De, A. (2015). Multiple imputation by fully conditional specification for dealing with missing data in a large epidemiologic study. *International Journal of Statistics in Medical Research*, 4, 287–295. http://dx.doi.org/10.6000/1929-6029.2015.04.03.7
- Lynn, P. (2009). Sample design for understanding society. Essex, England: University of Essex. https://econpapers.repec.org/paper/eseukhlsp/2009-01.htm
- Marshal, M. P., Dietz, L. J., Friedman, M. S., Stall, R., Smith, H. A., McGinley, J., . . . Brent, D. A. (2011). Suicidality and depression disparities between sexual minority and heterosexual youth: A metaanalytic review. *Journal of Adolescent Health*, 49, 115–123. http://dx .doi.org/10.1016/j.jadohealth.2011.02.005
- Marshal, M. P., Friedman, M. S., Stall, R., King, K. M., Miles, J., Gold, M. A., . . . Morse, J. Q. (2008). Sexual orientation and adolescent substance use: A meta-analysis and methodological review. *Addiction*, 103, 546–556. http://dx.doi.org/10.1111/j.1360-0443.2008.02149.x
- McCabe, C. J., Thomas, K. J., Brazier, J. E., & Coleman, P. (1996).Measuring the mental health status of a population: A comparison of the

- GHQ-12 and the SF-36 (MHI-5). *The British Journal of Psychiatry, 169*, 517–521. http://dx.doi.org/10.1192/bjp.169.4.516
- Menselson, T., Rehkopf, D. H., & Kubzansky, L. D. (2008). Depression among Latinos in the United States: A meta-analytic review. *Journal of Consulting and Clinical Psychology*, 76, 355–366. http://dx.doi.org/10 .1037/0022-006X.76.3.355
- Meyer, I. H. (2003). Prejudice, social stress, and mental health in lesbian, gay, and bisexual populations: Conceptual issues and research evidence. Psychological Bulletin, 129, 674–697. http://dx.doi.org/10.1037/0033-2909.129.5.674
- Missinne, S., & Bracke, P. (2012). Depressive symptoms among immigrants and ethnic minorities: A population-based study in 23 European countries. Social Psychiatry and Psychiatric Epidemiology, 47, 97–109. http://dx.doi.org/10.1007/s00127-010-0321-0
- Mood, C. (2010). Logistic Regression: Why we cannot do what we think we can do, and what we can do about it. *European Sociological Review*, 26, 67–82. http://dx.doi.org/10.1093/esr/jcp006
- Moradi, B., Wiseman, M. C., DeBlaere, C., Goodman, M. B., Sarkees, A., Brewster, M. E., & Huang, Y.-P. (2010). LGB of color and white individuals' perceptions of heterosexist stigma, internalized homophobia, and outness: Comparisons of levels and links. *The Counseling Psychologist*, 38, 397–424. http://dx.doi.org/10.1177/0011000 009335263
- Mustanski, B., Andrews, R., & Puckett, J. A. (2016). The effects of cumulative victimization on mental health among lesbian, gay, bisexual, and transgender adolescents and young adults. *American Journal of Public Health*, 106, 527–533. http://dx.doi.org/10.2105/AJPH.2015 302976
- Neighbors, H. W., Jackson, J. S., Campbell, L., & Williams, D. (1989). The influence of racial factors on psychiatric diagnosis: A review and suggestions for research. *Community Mental Health Journal*, 25, 301–311. http://dx.doi.org/10.1007/BF00755677
- Pachankis, J. E., & Goldfried, M. R. (2004). Clinical issues in working with lesbian, gay, and bisexual clients. *Psychotherapy: Theory, Research, Practice, Training, 41*, 227–246. http://dx.doi.org/10.1037/ 0033-3204.41.3.227
- Paradies, Y., Ben, J., Denson, N., Elias, A., Priest, N., Pieterse, A., . . . Gee, G. (2015). Racism as a determinant of health: A systematic review and meta-analysis. *PLoS ONE*, 10, e0138511. http://dx.doi.org/10.1371/journal.pone.0138511
- Pedersen, A. B., Mikkelsen, E. M., Cronin-Fenton, D., Kristensen, N. R., Pham, T. M., Pedersen, L., & Petersen, I. (2017). Missing data and multiple imputation in clinical epidemiological research. *Clinical Epidemiology*, 9, 157–166. http://dx.doi.org/10.2147/CLEP.S129785
- Plöderl, M., Sellmeier, M., Fartacek, C., Pichler, E.-M., Fartacek, R., & Kralovec, K. (2014). Explaining the suicide risk of sexual minority individuals by contrasting the minority stress model with suicide models. *Archives of Sexual Behavior*, 43, 1559–1570. http://dx.doi.org/10.1007/s10508-014-0268-4
- Plöderl, M., & Tremblay, P. (2015). Mental health of sexual minorities. A systematic review. *International Review of Psychiatry*, 27, 367–385. http://dx.doi.org/10.3109/09540261.2015.1083949
- Poteat, V. P., & Espelage, D. L. (2007). Predicting psychosocial consequences of homophobic victimization in middle school students. *The Journal of Early Adolescence*, 27, 175–191. http://dx.doi.org/10.1177/0272431606294839
- Priest, N., Walton, J., White, F., Kowal, E., Baker, A., & Paradies, Y. (2014). Understanding the complexities of ethnic-racial socialization processes for both minority and majority groups: A 30-year systematic review. *International Journal of Intercultural Relations*, 43(Part B), 139–155. http://dx.doi.org/10.1016/j.ijintrel.2014.08.003
- Reitzel, L. R., Smith, N. G., Obasi, E. M., Forney, M., & Leventhal, A. M. (2017). Perceived distress tolerance accounts for the covariance between discrimination experiences and anxiety symptoms among sexual minor-

- ity adults. *Journal of Anxiety Disorders*, 48, 22–27. http://dx.doi.org/10 .1016/j.janxdis.2016.07.006
- Rodriguez-Seijas, C., Eaton, N. R., & Pachankis, J. E. (2019). Prevalence of psychiatric disorders at the intersection of race and sexual orientation: Results from the National Epidemiologic Survey of Alcohol and Related Conditions-III. *Journal of Consulting and Clinical Psychology*, 87, 321–331. http://dx.doi.org/10.1037/ccp0000377
- Rosario, M., Schrimshaw, E. W., Hunter, J., & Gwadz, M. (2002). Gay-related stress and emotional distress among gay, lesbian, and bisexual youths: A longitudinal examination. *Journal of Consulting and Clinical Psychology*, 70, 967–975. http://dx.doi.org/10.1037/0022-006X.70.4 967
- Russell, S. T., & Truon, N. L. (2001). Adolescent sexual orientation, race and ethnicity, and school environments: A national study of sexual minority youth of color. In K. K. Kumashiro (Ed.), *Troubling intersec*tions of race and sexuality (pp. 113–130). Lanham, MD: Rowman & Littlefield.
- Schafer, J. L., & Graham, J. W. (2002). Missing data: Our view of the state of the art. *Psychological Methods*, 7, 147–177. http://dx.doi.org/10 .1037/1082-989X.7.2.147
- Shevlin, M., & Adamson, G. (2005). Alternative factor models and factorial invariance of the GHQ-12: A large sample analysis using confirmatory factor analysis. *Psychological Assessment*, 17, 231–236. http://dx.doi.org/10.1037/1040-3590.17.2.231

- Timmins, L., Rimes, K. A., & Rahman, Q. (2020). Minority stressors, rumination, and psychological distress in lesbian, gay, and bisexual individuals. *Archives of Sexual Behavior*, 49, 661–680. http://dx.doi.org/10.1007/s10508-019-01502-2
- Vanden Berghe, W., Dewaele, A., Cox, N., & Vincke, J. (2010). Minority specific determinants of mental well-being among lesbian, gay, and bisexual youth. *Journal of Applied Social Psychology*, 40, 153–166. http://dx.doi.org/10.1111/j.1559-1816.2009.00567.x
- Vaughn, A. A., Roesch, S. C., & Aldridge, A. A. (2009). Stress-related growth in racial/ethnic minority adolescents: Measurement structure and validity. *Educational and Psychological Measurement*, 69, 131–145. http://dx.doi.org/10.1177/0013164408318775
- Velez, B. L., Watson, L. B., Cox, R., & Flores, M. J. (2017). Minority stress and racial or ethnic minority status: A test of the greater risk perspective. *Psychology of Sexual Orientation and Gender Diversity*, 4, 257–271. http://dx.doi.org/10.1037/sgd0000226
- Vertovec, S. (2007). Super-diversity and its implications. *Ethnic and Racial Studies*, 30, 1024–1054. http://dx.doi.org/10.1080/014198 70701599465

Received August 29, 2019
Revision received May 21, 2020
Accepted May 28, 2020